Author's response to reviews

Title: Do patients with recurrent reported episodes of campylobacteriosis differ from those with a single disease event?

Authors:

Julie Arsenault (julie.arsenault@umontreal.ca)
André Ravel (andre.ravel@phac-aspc.gc.ca)
Pascal Michel (pascal.michel@phac-aspc.gc.ca)
Olaf Berke (oberke@uoguelph.ca)
Pierre Gosselin (pierre.gosselin@inpq.qc.ca)

Version: 4 Date: 11 January 2011

Author's response to reviews:

Dear editor,

We made a few editorial changes to our manuscript, as described below:

1. Page 1, Email addresses: "andre.Ravel" was changed to "andre.ravel".
2. Page 10 line 8: a missing parenthesis was added after "(Table 1".
3. Page 10 line 15: (P<0.0001, LRT) was changed to "(P<0.001, LRT)"
4. Page 12 line 21: an extra space was removed between reference and period ("[21].")
5. Page 13 line 5: the reference "[21] [23-25]" was changed to "[21, 23-25]"
6. Page 13 lines 9 and 15: an extra space was removed between reference and period ("[29]." and "[31].")
9. Page 23, Figure 1. "Bottom: Cases" was changed to "Bottom: cases"

Best regards,
Julie Arsenault